- Check for updates
- ¹ VA Northern California Healthcare System, CA, USA
- ² University of California, Davis, CA, USA
- ³ VA Palo Alto Healthcare System, Palo Alto, CA, USA
- ⁴ Stanford University, Stanford, CA, USA

Correspondence to: V Manja veena.manja@va.gov Cite this as: *BMJ* 2024;387:q2424 http://dx.doi.org/10.1136/bmj.q2424 Safety of sodium-glucose cotransporter-2 inhibitors for heart failure

Drawing inferences from observational data with possible confounding

Veena Manja, ^{1, 2} Paul Heidenreich^{3, 4}

Heart failure is increasing in prevalence and is a major cause of morbidity and mortality worldwide,¹ with prevalence ranging from 1% to 3% of the general adult population in high income countries.¹ Limited data from low and middle income countries suggest high heart failure disease burden.²³ Heart failure with reduced ejection fraction, defined as a left ventricular ejection fraction of $\leq 40\%$, accounts for around 30-60% of heart failure in epidemiological studies.¹ Increasing evidence of the effectiveness of certain drugs to reduce mortality and morbidity in heart failure with reduced ejection fraction has led to strong recommendations for their use in clinical practice guidelines.⁴⁵ The foundational therapeutic agents for heart failure with reduced ejection fraction have been shown to improve survival, reduce the risk of readmission to hospital, and improve symptoms by targeting the renin-angiotensin-aldosterone and sympathetic nervous systems. These include angiotensin converting enzyme inhibitors, angiotensin receptor blockers, angiotensin receptor-neprilysin inhibitors, mineralocorticoid receptor antagonists, and ß blockers. In recent randomized controlled trials, the addition of a sodium-glucose cotransporter-2 (SGLT-2) inhibitor further reduced the risk of worsening heart failure and death from cardiovascular disease in patients with heart failure with reduced ejection fraction.⁴⁵ In the linked study (doi:10.1136/bmj-2024-080925), Svanström and colleagues add to this growing evidence by using real world administrative data to show a reduction in mortality but no change in heart failure related hospital admissions with SGLT-2 inhibitor use.⁶

The authors linked data in the Danish heart failure registry to the national civil registration system, including data for patients aged \geq 45 years with a left ventricular ejection fraction of ≤40% treated from July 2020 to June 2023. They used a modified prevalent new user design,⁷ with an intervention group including patients starting SGLT-2 inhibitor treatment for a heart failure indication and a comparator group including patients without SGLT-2 inhibitors matched on time since diagnosis of heart failure. Results were adjusted using inverse probability of treatment weighting to account for differences in baseline characteristics. The primary outcome was all cause mortality, and secondary outcomes were a composite of cardiovascular mortality or hospital admission with heart failure and its components. The authors used proportional hazards regression to compare outcomes in the intervention and comparator groups. They report a 25% relative risk reduction for all cause mortality and a 23% reduction in cardiovascular mortality but no change in the composite of cardiovascular mortality

or hospital admissions due to heart failure associated with SGLT-2 inhibitor use compared with non-users. The magnitude of association with reduction in mortality was consistent across subgroups. They do not report non-heart failure related or all cause hospital admisstions.

Given that observational data on treatment effectiveness are often confounded in ways that cannot be eliminated through risk adjustment, one must be careful in drawing conclusions.⁸ Observational data can be useful in examining outcomes or subgroups that were too small to be adequately evaluated in randomized trials. If investigators can first show that the observational outcome is similar to that observed in similar patients in randomized trials, confidence in the observed result will be greater for other populations or other outcomes than for studies in which the investigators cannot reproduce the results of the clinical trials. If the clinical trial results cannot be reproduced, one must have a strong biological plausibility for why the observational studies, with their risk of confounding, are more accurate than the clinical trial results.

Unfortunately, the clinical trial data differ, as noted by Svanström and colleagues. In a meta-analysis of randomized trials of SGLT-2 inhibitors in patients with heart failure with reduced ejection fraction, all cause mortality was reduced with an odds ratio of 0.87 (95% confidence interval 0.77 to 0.98), an effect size half that observed in the linked study.⁹ By contrast, hospital admission was markedly reduced in the clinical trials (odds ratio 0.69, 0.62 to 0.78) but not in the registry. How does one reconcile these differences between the randomized controlled trials and observational studies? The authors suggest that their reliance on coding of heart failure for assigning a hospital admission due to heart failure may explain their lack of reduced admissions. Although coding is inferior to adjudication using the medical record, accuracy of coding would have to have been much poorer than has been reported to account for all of the difference.¹⁰ Another possible explanation is that patients not treated with SGLT-2 inhibitors may have had non-heart failure disease that was more severe than their heart failure whereas those treated had heart failure as the major condition. Although the investigators were able to match baseline demographic characteristics between the groups, the patients treated with SGLT-2 inhibitors may be different from those not treated in ways that were not considered or measured but that affect mortality, such as frailty. If risk adjustment was incomplete then these non-treated patients would have worse mortality (from non-cardiovascular causes) and be less likely to be admitted to hospital for heart failure

EDITORIALS

(and presumably more likely to be admitted for non-heart failure conditions).

In spite of these limitations, these results provide assurance that no unexpected harm results from SGLT-2 inhibitors when they are used for treatment of heart failure outside the clinical trial setting. In addition, the large number of patients treated in the registry shows the rapid uptake of this fourth pillar of pharmacotherapy for heart failure with reduced ejection fraction.⁴⁵ However, SGLT-2 inhibitors are still underused.¹¹ Among practitioners, the potential for euglycemic diabetic ketoacidosis, complexity of patients, and drug costs may lead to hesitancy to prescribe SGLT-2 inhibitors.¹² Robust implementation efforts should tackle barriers to prescribing in an effort to increase evidence based prescribing.

Competing interests: The BMJ has judged that there are no disqualifying financial ties to commercial companies. The authors declare the following other interests: none. Further details of The BMJ policy on financial interests is here: https://www.bmj.com/sites/default/files/attachments/resources/2016/03/16-current-bmj-education-coi-form.pdf

Provenance and peer review: commissioned, not externally peer reviewed

- Savarese G, Becher PM, Lund LH, Seferovic P, Rosano GMC, Coats AJS. Global burden of heart failure: a comprehensive and updated review of epidemiology. *Cardiovasc Res* 2023;118:-87. doi: 10.1093/cvr/cvac013 pmid: 35150240
- 2 Callender T, Woodward M, Roth G, etal. Heart failure care in low- and middle-income countries: a systematic review and meta-analysis. *PLoS Med* 2014;11:e1001699. doi: 10.1371/journal.pmed.1001699 pmid: 25117081
- ³ Dokainish H, Teo K, Zhu J, etal. Heart failure in low- and middle-income countries: background, rationale, and design of the INTERnational Congestive Heart Failure Study (INTER-CHF). Am Heart /2015;170:-634.e1. doi: 10.1016/j.ahj.2015.07.008 pmid: 26386785
- 4 Heidenreich PA, Bozkurt B, Aguilar D, etal. 2022 AHA/ACC/HFSA Guideline for the Management of Heart Failure: A Report of the American College of Cardiology/American Heart Association Joint Committee on Clinical Practice Guidelines. J Am Coll Cardiol 2022;79:-421. doi: 10.1016/j.jacc.2021.12.012 pmid: 35379503
- 5 McDonagh TA, Metra M, Adamo M, etal. Corrigendum to: 2021 ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure: Developed by the Task Force for the diagnosis and treatment of acute and chronic heart failure of the European Society of Cardiology (ESC) With the special contribution of the Heart Failure Association (HFA) of the ESC. *Eur Heart J* 2021;42:. doi: 10.1093/eurheartj/ehab670 pmid: 34649282
- 6 Svanström H, Mkoma GF, Hviid A, Pasternak B. SGLT2 inhibitors and mortality among patients with heart failure with reduced ejection fraction: linked database study. *BMJ* 2024;387:e080925.
- 7 Suissa S, Dell'Aniello S, Renoux C. The Prevalent New-user Design for Studies With no Active Comparator: The Example of Statins and Cancer. *Epidemiology* 2023;34:-9. doi: 10.1097/EDE.00000000001628 pmid: 37195285
- 8 Freemantle N, Marston L, Walters K, Wood J, Reynolds MR, Petersen I. Making inferences on treatment effects from real world data: propensity scores, confounding by indication, and other perils for the unwary in observational research. *BMJ* 2013;347:. doi: 10.1136/bmj.f6409 pmid: 24217206
- 9 Vaduganathan M, Docherty KF, Claggett BL, etal. SGLT-2 inhibitors in patients with heart failure: a comprehensive meta-analysis of five randomised controlled trials. *Lancet* 2022;400:-67. doi: 10.1016/S0140-6736(22)01429-5 pmid: 36041474
- Sandhu AT, Zheng J, Skye M, Heidenreich PA. Updating the Accuracy of Administrative Claims for Identifying Left Ventricular Ejection Fraction Among Patients With Heart Failure. *Circ Cardiovasc Qual Outcomes* 2023;16:e008919. doi: 10.1161/CIRCOUTCOMES.122.008919 pmid: 36924223
- 11 Chakrala T, Prakash RO, Kim J, etal. Prescribing patterns of SGLT-2 inhibitors for patients with heart failure: A two-center analysis. *Am Heart J Plus* 2023;28:100286. doi: 10.1016/j.ahjo.2023.100286 pmid: 38511072
- 12 Teng LE, Lammoza N, Aung AK, etal. General physicians' perspectives on SGLT2 inhibitors for heart failure. *Intern Med* J 2024;54:-9. doi: 10.1111/imj.16440 pmid: 38957943